Author’s response to reviews

Title: Work-family Balance and the Subjective Well-being of Rural Women in Sichuan, China

Authors:

Yue Shui (shuiyue16@imde.ac.cn)

Dingde Xu (dingdexu@126.com)

Yi Liu (18583287370@163.com)

Shaoquan Liu (liushq@imde.ac.cn)

Version: 3 Date: 06 Nov 2019

Author’s response to reviews:

Reviewer reports:

Abqariyah Yahya (Reviewer 2):

Dear author, Thank you for providing new analysis considering the weightage of the samples. Unfortunately, I could not find the description on how the weightage was used in the analysis.

Kindly provide a concise description on how you deal with the weightage in your analysis and present both the unweighted and complex sample analysis (weighted) side-by-side for better comparison. You may also discuss the advantages of conducting and presenting complex sample analysis.

Response: Thank you very much for your helpful suggestion. I made the following changes according to your suggestion: I added the description on how the weightage was used in the analysis in 2.3 Econometric model (line 220-223, page 12). At the same time, I put the analysis results before and after adding the weight into the table 3(line 320-322, page 18-19). In the description section, it also explains the advantage of conducting and presenting complex sample analysis (after adding weights, the model has a better goodness of fit). (Line 249-269, page 14-15)